Author’s response to reviews

Title: Implementing Blended Learning in Emergency Airway Management Training: A Randomized Controlled Trial

Authors:

Madeleine Huei Tze Kho (madeleine.kho.ht@gmail.com)
Keng Sheng Chew (cksheng74@yahoo.com)
Muhamin Noor Azhar (muhamin@um.edu.my)
Mohd Lotfi Hamzah (drlogp@yahoo.com)
Kee Man Chuah (kmchuah@unimas.my)
Aida Bustam (aiibustam@gmail.com)
Hiang Chuan Chan (hcc2000us@yahoo.com)

Version: 2 Date: 22 Dec 2017

Author’s response to reviews:

1. Please kindly provide all authors emails on title page.
Response: Done

2. Please include a "Trial Registration" at the end of Abstract: listing the trial registry and the unique identifying number and the date of registration, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Registered 28 September 2004. Please note that there should be no space between the letters and numbers of the trial registration number.
Response: Done. The trial registration is now included in lines 58-59

3. Please provide us a clean final version without track changes.
Response: Done